Author’s response to reviews

Title: The Trend and Features of Physician Workforce Supply in China: After National Medical Licensing System Reform

Authors:
Chengxiang Tang (tang.chengxiang@gmail.com)
Daisheng Tang (dstang@bjtu.edu.cn)

Version: 4 Date: 25 Feb 2018

Author's response to reviews:

***Thank you very much for your encouragement and pointing out an advantage of this work. We further improve this manuscript by following your advice: first, we amended the Method section by adding a paragraph instead of inserting the footnote in page 8. The WHO and ISCO two references were appended as an integral part of the list of references at the end of article.

Second, it should be clear that the “(Netherlands, 2014)” refers to in-text citation of the ISCO reference. I apologize for any confusion made in last response.